Author's response to reviews

Title: Risk factors for latent tuberculosis infection in close contacts of active tuberculosis patients in South Korea: a prospective cohort study

Authors:

Seung Jun Lee (juny2278@naver.com)
Seung Hun Lee (bs1974@nate.com)
You Eun Kim (narr0707@naver.com)
Yu Ji Cho (im-ug@hanmail.net)
Yi Yeong Jeong (dr202202@naver.com)
Jong Deog Lee (ljd@gnu.ac.kr)
Jang Rak Kim (jrkim@gnu.ac.kr)
YoungSil Hwang (yshwang@gnu.ac.kr)
HeeJin Kim (hatchwingbird@yahoo.co.kr)
Richard Menzies (dick.menzies@mcgill.ca)

Version: 2 Date: 5 August 2013

Author's response to reviews:

Dear editors,

Revised text details are as follows;

1. The name and number of ethics committee that approved this study are included in Methods section.
2. Funding statement was moved from Competing Interest section to Acknowledgement section.

Sincerely,